Author's response to reviews

Title: Design and validation of the Health Professionals' Attitudes Toward the Homeless Inventory (HPATHI)

Authors:

David S. Buck (dbuck@bcm.tmc.edu)
F. MARCONI Monteiro (mntrmjmonteiro@aol.com)
Suzanne Kneuper (skneuper@bcm.tmc.edu)
Donna Rochon (drochon@bcm.tmc.edu)
Dana L. Clark (dlclark@bcm.tmc.edu)
Allegra Melillo (allemeli@hotmail.com)
Robert J. Volk (bvolk@bcm.tmc.edu)

Version: 3 Date: 7 January 2005

Author's response to reviews:

Changes to HPATHI manuscript

1. The title in the manuscript and the submission system has been changed to sentence case.
2. The footer has been removed from the manuscript.
3. The abstract has been rephrased and the first two sentences have been inverted.
4. Sentence case has been used for all headings and sub-headings in the manuscript.
5. As per the reviewer's recommendation, "affects" has been changed to "correlates with" in the first sentence of the second paragraph in the Discussion section.